Author’s response to reviews

Title: Pre-operative Biomarkers and Imaging Tests as Predictors of Post-operative Delirium in Non-cardiac Surgical Patients: A Systematic Review

Authors:

Farrah Ayob (farrah187@icloud.com)
Enoch Lam (Elam85@uwo.ca)
George Ho (ho.george128@gmail.com)
Frances Chung (frances.chung@uhn.ca)
Hossam El-Beheiry (Hossam.El-Beheiry@thp.ca)
Jean Wong (jean.wong@uhn.ca)

Version: 3 Date: 24 Jan 2019

Author’s response to reviews:

Dear Editor/ Reviewers,

Re: "Pre-operative Biomarkers and Imaging Tests as Predictors of Post-operative Delirium in Non-cardiac Surgical Patients: A Systematic Review" (BANE-D-18-00347R2)

Thank you for your comments. Further to the recommendations, we have made the changes:

1. Funding – We have stated the ‘Department of Anesthesia, Toronto Western Hospital, University Health Network’ as our funder and that the department had no role in the design of the study.

2. Additional files: N/A.

3. Figure legends were removed from figure files.
4. A clean manuscript is submitted.

Best wishes,

Farrah Ayob
Jean Wong